Author’s response to reviews

Title: Interaction between COX-1 and COX-2 increases susceptibility to ischemic stroke in a Chinese population

Authors:

Lei Zhao (zhaolei127@qq.com)
Jinghuan Fang (121208218@qq.com)
Muke Zhou (zmkemail@126.com)
Jie Zhou (452659940@qq.com)
Lihua Yu (498173816@qq.com)
Ning Chen (jingning22@163.com)
Li He (heli2003new@126.com)

Version: 1 Date: 23 Jan 2019

Author’s response to reviews:

comments:

1. Please amend your manuscript so that your consent to participate statements are consistent and correct. Please also specify whether consent was given in written or verbal form.

   We have amend our statement so that our consent to participate statement are consistent and correct. We have revised our statement as“Our research was reviewed and approved by the Ethics Review Committee of West China hospital, Sichuan University. The patient consent was given in verbal form”.

2. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

   We have added all the authors' email on the title page.

Thank you for your help.